Title: Successful treatment of Rhodesian trypanosomiasis in a Polish tourist returning from Uganda and Rwanda: a case report.

Authors:

Malgorzata Paul (mpaul@ump.edu.pl)
Jerzy Stefaniak (medtrop@mp.pl)
Leon Drobnik (tropiki@spsk2.pl)
Marjan Esbroeck Van (jclerinx@itg.be)
Dirk Geysen (jclerinx@itg.be)
Jan Clerinx (jclerinx@itg.be)

Version: 2 Date: 27 August 2013

Author's response to reviews: see over
Dear Editor,

I enclose a new version of a manuscript No 636790154106 1967 for a consideration for publication in the BMC Infectious Diseases with some corrections according to suggestions of the Journal Editorial Office. The editorial changes which have been made are as follows:

1) An information about treatment “with pentamidine” has been removed from the title, so the final version of the title is “Successful treatment of Rhodesian trypanosomiasis in a Polish tourist returning from Uganda and Rwanda: a case report”;

2) Separate e-mail addresses have been provided for each of co-authors from the Swiss Tropical and Public Health Institute in Basel (Belgium), and two missing e-mail addresses have been added in the authors’ section: mvesbroeck@itg.be for Marjan Van Esbroeck, and dgeysen@itg.be for Dirk Geysen.

The proposed paper has not been published before nor sent to another journal for evaluation. In the manuscript, we present an exceptional and the first case of severe African sleeping sickness imported to Poland and Central-Eastern Europe by a patient returning from Uganda, and which was successfully treated with pentamidine. This is probably the first case of imported severe T. brucei rhodesiense trypanosomiasis sensitive to pentamidine so far described in literature.

As corresponding author, I would be grateful if you could address any queries to me at the address/email address given below.

Yours sincerely,

Malgorzata Paul, MD, PhD, DSc, DTM&H
Adjunct Professor
Department and Clinic of Tropical and Parasitic Diseases
Poznan University of Medical Sciences
Przybyszewskiego Street 49
60-355 Poznan
Poland
phone: +48 618691363
fax: +48 618691699
e-mail: mpaul@ump.edu.pl

Poznan, August 27, 2013